Author's response to reviews

Title: Assessing Disease Disclosure in Adults with Cystic Fibrosis: The Adult Data for Understanding Lifestyle and Transitions Survey

Authors:

Avani C Modi (avani.modi@cchmc.org)
Alexandra L Quittner (aquittner@miami.edu)
Michael P Boyle (mboyle@jhmi.edu)

Version: 3 Date: 2 September 2010

Author's response to reviews: see over
Dear Dr. Shipley,

We are excited that our paper has been accepted for publication, pending clarification about human subjects approval. I have added the following statement into the paper in the procedures section regarding this issue: “All authors sent the current protocol to their respective Institutional Review Boards, with each IRB deeming that approval was unnecessary due to the de-identified nature of the anonymous survey. Consent to participate was implicit to adult participants based on completion of the survey.” Please let me know if you have any questions regarding this issue.

Sincerely,

Avani Modi, Ph.D.
Assistant Professor of Pediatrics
Cincinnati Children’s Hospital Medical Center
Division of Behavioral Medicine and Clinical Psychology
3333 Burnet Avenue – MLC 7039 (Adherence Center)
Cincinnati, OH 45229
(513)636-4864; avani.modi@cchmc.org